Author’s response to reviews

Title: MEG3 promotes proliferation and inhibits apoptosis in osteoarthritis chondrocytes by miR-361-5p/FOXO1 axis

Authors:

Anying Wang (wanganying008@163.com)
Naixia Hu (15854858101@163.com)
Yefeng Zhang (zhangyefeng3395@sohu.com)
Yuanzhen Chen (chenyuanzhen2011@163.com)
Changhui Su (suchanghui821325@163.com)
Yao Lv (574026615@qq.com)
Yong Shen (shenyong2483@163.com)

Version: 5 Date: 17 Dec 2019

Author’s response to reviews:

MGNM-D-19-00237R4MEG3 promotes proliferation and inhibits apoptosis in osteoarthritis chondrocytes by miR-361-5p/FOXO1 axis

Dear Editor,

Thank you very much for your letter and advice. We have revised the manuscript, and would like to re-submit it for your consideration. We have addressed the comments raised by the reviewers, and the amendments are highlighted in red in the revised manuscript. Point by point responses to the reviewers’ comments are listed below this letter. This manuscript has been edited and proofread by a native English language speaker. We hope that the revised version of the manuscript is now acceptable for publication in your journal. In addition, the name of organization 3 to which the author Naixia Hu belongs has changed. So I made a change in the title page. I hope you'll allow me to do so. Thanks. I look forward to hearing from you soon.

With best wishes,

Yours sincerely,

Yong Shen.
available from the corresponding author on reasonable request. 2. Copy edit I'm afraid the quality of the English used throughout your manuscript still doesn't meet our requirements. Please use a professional language editing service to copy edit your manuscript. If you wish to have the language in your manuscript edited by a native-English speaker with scientific expertise, BioMed Central has a new in house editing service. The new editing tool can provide both scientific and language editing: http://authorservices.springernature.com/. If you use a professional editing service, please ensure that you attach the copy edit certificate from the professional editing service that you used as an additional file. Please also note that use of a professional language service is not a guarantee of acceptance for publication. Answer: Based on your suggestion, we have edited this article using a professional editing service and corresponding corrections have been made in the revised version. The copy edit certificate from the professional editing service has been provided as an additional file. 3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Should you wish to respond to these revision requests, please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. Answer: I have uploaded these files as required.